Author's response to reviews

Title: Successful C1 inhibitor short-term prophylaxis during redo mitral valve replacement in a patient with hereditary angioedema

Authors:

Jonathan Bernstein (jonathan.bernstein@uc.edu)
Suzanne Coleman (suzanne.coleman@khnetwork.org)
Arturo J Bonnin (ajbonnin@gmail.com)

Version: 5 Date: 15 October 2010

Author's response to reviews:

The revisions are to remove information about attempts to measure C1INH and HMWkininogen before during and after treatment and to remove acknowledgement of Bruce Zuraw from the end of the paper.